\documentclass[12pt]{article}
\usepackage{e-jc}
\specs{P45}{20(3)}{2013}

% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need

% we recommend these ams packages
\usepackage{amsthm,amsmath,amssymb}

% we recommend the graphicx package for importing figures
\usepackage{graphicx}

% use this command to create hyperlinks (optional and recommended)
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

% use these commands for typesetting doi and arXiv references in the bibliography
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

% all overfull boxes must be fixed; 
% i.e. there must be no text protruding into the margins


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% if needed include a line break (\\) at an appropriate place in the title


\title{\bf Delay colourings of cubic graphs}

\author{Agelos Georgakopoulos\\
\small   {Mathematics Institute}\\[-0.8ex]
\small    {University of Warwick}\\[-0.8ex]
\small   {CV4 7AL, UK}\\}



% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{\dateline{Nov 24, 2012}{Aug 29, 2013}{Sep 20, 2013}\\
\small Mathematics Subject Classifications: 05C15, 05C70}


\newcommand{\COLORON}{0}
\newcommand{\NOTESON}{1}
\newcommand{\Debug}{0}


%&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&
\usepackage[usenames,dvips]{color} 
\usepackage{amsthm,amssymb,amsmath,enumerate,graphicx,epsf,stmaryrd}

\hyphenation{com-pac-ti-fi-cation}

\newcommand{\keywordname}{{\bf Keywords:}}
\newcommand{\keywords}[1]{\par\addvspace\baselineskip
	\noindent\keywordname\enspace\ignorespaces#1}
	
\renewcommand{\theenumi}{(\roman{enumi})}
\renewcommand{\labelenumi}{\theenumi}

\newcommand{\comment}[1]{}
\newcommand{\COMMENT}[1]{}

\definecolor{darkgray}{rgb}{0.3,0.3,0.3}
\newcommand{\defi}[1]{{\color{darkgray}\emph{#1}}}

\newcommand{\acknowledgement}{\section*{Acknowledgement}}
\newcommand{\acknowledgements}{\section*{Acknowledgements}}

%--------------------
% USEFUL STRUCTURES
%--------------------

%\newcommand{}{}

\comment{
	\begin{lemma}\label{}	
\end{lemma}
% *** ---- *** 
\begin{proof}

\end{proof}


\begin{theorem}\label{}
\end{theorem} 
% *** ---- *** 
\begin{proof} 	

\end{proof}

%\begin{conjecture}\label{}\end{conjecture} 
%\begin{corollary}\label{}\end{corollary} % *** ---- *** \begin{proof} 	\end{proof}
%\begin{example}\label{}\end{example} % *** ---- *** \begin{proof} 	\end{proof}
}

% THIS CREATES A PARAGRAPH WITH A (*) NEXT TO IT THAT YOU CAN REFER TO
%\begin{equation} \label{star} \begin{minipage}[c]{0.85\textwidth}
% bla
%\end{minipage}\ignorespacesafterend \tag{\ensuremath{*}} \end{equation}


%----------------------
% THEOREMS
%----------------------

\newtheorem{proposition}{Proposition}[section]
\newtheorem{definition}[proposition]{Definition}
\newtheorem{theorem}[proposition]{Theorem}
\newtheorem{corollary}[proposition]{Corollary}
\newtheorem*{nocorollary}{Corollary}
\newtheorem{lemma}[proposition]{Lemma}
\newtheorem{observation}[proposition]{Observation}
\newtheorem{conjecture}{{\color{red}Conjecture}}[section]
\newtheorem*{noConjecture}{{\color{red}Con\-jecture}}
\newtheorem{problem}[conjecture]{{\color{red}Problem}}
\newtheorem*{noProblem}{{\color{red}Problem}}
\newtheorem*{noTheorem}{Theorem}
\newtheorem{examp}[proposition]{Example}%[section]
\newtheorem{claim}{Claim}
\newtheorem*{noclaim}{Claim}

% GERMAN
\newtheorem{vermutung}{{\color{red}Vermutung}}[section]
\newtheorem*{noVermutung}{{\color{red}Vermutung}}


\newcommand{\example}[2]{\begin{examp} \label{#1} {{#2}}\end{examp}}

%\newcommand{\obda}{\textrm{w.l.o.g. }}
\newcommand{\kreis}[1]{\mathaccent"7017\relax #1}

\newcommand{\FIG}{0}

\ifnum \NOTESON = 1 \newcommand{\note}[1]{ 

\hspace*{-30pt}
	{\color{blue}  NOTE: \color{Turquoise}{\small  \tt \begin{minipage}[c]{1.1\textwidth}  #1 \end{minipage} \ignorespacesafterend }} 
	
	}
\else \newcommand{\note}[1]{} \fi

\newcommand{\afsubm}[1]{ \ifnum \Debug = 1 {\mymargin{#1}}
\fi} %For notes on after-submission changes

\ifnum \Debug = 1 \newcommand{\sss}{\ensuremath{\color{red} \bowtie \bowtie \bowtie\ }}
\else \newcommand{\sss}{} \fi

\ifnum \FIG = 1 \newcommand{\fig}[1]{Figure ``{#1}''}
\else \newcommand{\fig}[1]{Figure~\ref{#1}} \fi

\ifnum \FIG = 1 \newcommand{\figs}[1]{Figures ``{#1}''}
\else \newcommand{\figs}[1]{Figures~\ref{#1}} \fi

\ifnum \Debug = 1 \usepackage[notref,notcite]{showkeys}
\fi

\ifnum \COLORON = 0 \renewcommand{\color}[1]{}
\fi

\newcommand{\Fig}{Figure}


%   \epsfxsize=0.8\hsize
\newcommand{\showFigu}[3]{
   \begin{figure}[htbp]
   \centering
   \noindent
   \epsfxsize=0.8\hsize
   \epsfbox{#1.eps}
%   \includegraphics[width=0.8\linewidth]{#1}
   \caption{\small #3}
   \label{#2}
   \end{figure}
}

\newcommand{\showFig}[2]{
   \begin{figure}[htbp]
   \centering
   \noindent
   \epsfbox{#1.eps}
   \caption{\small #2}
   \label{#1}
   \end{figure}
}

\newcommand{\showFigI}[1]{ % no caption
   \begin{figure}[htbp]
   \centering
   \noindent
   \epsfbox{#1.eps}
%   \caption{}
   \label{#1}
   \end{figure}
}



%--------------------
% LETTERS
%--------------------

\newcommand{\N}{\ensuremath{\mathbb N}}
\newcommand{\R}{\ensuremath{\mathbb R}}
\newcommand{\Z}{\ensuremath{\mathbb Z}}
\newcommand{\Q}{\ensuremath{\mathbb Q}}

\newcommand{\ca}{\ensuremath{\mathcal A}}
\newcommand{\cb}{\ensuremath{\mathcal B}}
\newcommand{\cc}{\ensuremath{\mathcal C}}
\newcommand{\cd}{\ensuremath{\mathcal D}}
\newcommand{\ce}{\ensuremath{\mathcal E}}
\newcommand{\cf}{\ensuremath{\mathcal F}}
\newcommand{\cg}{\ensuremath{\mathcal G}}
\newcommand{\ch}{\ensuremath{\mathcal H}}
\newcommand{\ci}{\ensuremath{\mathcal I}}
\newcommand{\cj}{\ensuremath{\mathcal J}}
\newcommand{\ck}{\ensuremath{\mathcal K}}
\newcommand{\cl}{\ensuremath{\mathcal L}}
\newcommand{\cm}{\ensuremath{\mathcal M}}
\newcommand{\cn}{\ensuremath{\mathcal N}}
\newcommand{\co}{\ensuremath{\mathcal O}}
\newcommand{\cp}{\ensuremath{\mathcal P}}
\newcommand{\cq}{\ensuremath{\mathcal Q}}
\newcommand{\cs}{\ensuremath{\mathcal S}}
\newcommand{\cgr}{\ensuremath{\mathcal R}}
\newcommand{\ct}{\ensuremath{\mathcal T}}
\newcommand{\cu}{\ensuremath{\mathcal U}}
\newcommand{\cv}{\ensuremath{\mathcal V}}
\newcommand{\cx}{\ensuremath{\mathcal X}}
\newcommand{\cy}{\ensuremath{\mathcal Y}}
\newcommand{\cz}{\ensuremath{\mathcal Z}}
\newcommand{\cw}{\ensuremath{\mathcal W}}

\newcommand{\oo}{\ensuremath{\omega}}
\newcommand{\OO}{\ensuremath{\Omega}}
\newcommand{\alp}{\ensuremath{\alpha}}
\newcommand{\bet}{\ensuremath{\beta}}
\newcommand{\gam}{\ensuremath{\gamma}}
\newcommand{\Gam}{\ensuremath{\Gamma}}
\newcommand{\del}{\ensuremath{\delta}}
\newcommand{\eps}{\ensuremath{\epsilon}}
\newcommand{\kap}{\ensuremath{\kappa}}
\newcommand{\lam}{\ensuremath{\lambda}}
\newcommand{\sig}{\ensuremath{\sigma}}
\newcommand{\al}{\ensuremath{\alpha}}
\newcommand{\el}{\ensuremath{\ell}}


\newcommand{\CCC}{\ensuremath{\mathcal C}}
\newcommand{\ccc}{\ensuremath{\mathcal C}}
\newcommand{\ccg}{\ensuremath{\mathcal C(G)}}
\newcommand{\fcg}{\ensuremath{|G|}}

%--------------------
% SYMBOLS
%--------------------

\newcommand{\zero}{\mathbb 0}
\newcommand{\sm}{\backslash}
\newcommand{\restr}{\upharpoonright}
\newcommand{\noproof}{\qed}
\newcommand{\isom}{\cong}
\newcommand{\subgr}{\subseteq}

%--------------------
% MISC
%--------------------


\newcommand{\ztwo}{\mathbb Z_2}
\newcommand{\nin}{\ensuremath{{n\in\N}}}
\newcommand{\iin}{\ensuremath{{i\in\N}}}
\newcommand{\unin}{\ensuremath{[0,1]}}
\newcommand{\xy}{$x$-$y$}

\newcommand{\limf}[1]{\ensuremath{\lim \inf(#1)}}


\newcommand{\sgl}[1]{\ensuremath{\{#1\}}}
\newcommand{\pth}[2]{\ensuremath{#1}\text{--}\ensuremath{#2}~path}
\newcommand{\topth}[2]{topological \ensuremath{#1}\text{--}\ensuremath{#2}~path}
\newcommand{\topths}[2]{topological 
	\ensuremath{#1}\text{--}\ensuremath{#2}~paths}
\newcommand{\pths}[2]{\ensuremath{#1}\text{--}\ensuremath{#2}~paths}
\newcommand{\arc}[2]{\ensuremath{#1}\text{--}\ensuremath{#2}~arc}
\newcommand{\arcs}[2]{\ensuremath{#1}\text{--}\ensuremath{#2}~arcs}
\newcommand{\seq}[1]{\ensuremath{(#1_n)_{n\in\N}}} 
\newcommand{\sseq}[2]{\ensuremath{(#1_i)_{i\in #2}}} %(sub-)sequence with a given index set
\newcommand{\susq}[2]{\ensuremath{(#1_{#2_i})_{\iin}}} %(sub-)sequence with subindex
\newcommand{\oseq}[2]{\ensuremath{(#1_\alp)_{\alp< #2}}} %ordinal-indexed sequence
\newcommand{\fml}[1]{\ensuremath{\{#1_i\}_{i\in I}}} 
\newcommand{\ffml}[2]{\ensuremath{\{#1_i\}_{i\in #2}}} % family with given index set
\newcommand{\ofml}[2]{\ensuremath{\{#1_\alp\}_{\alp< #2}}} % family indexed by ordinal
%\newcommand{\fam}[1]{\fml{#1}} 
\newcommand{\flo}[2]{\ensuremath{#1}\text{--}\ensuremath{#2}~flow} % x-y flow
\newcommand{\flos}[2]{\ensuremath{#1}\text{--}\ensuremath{#2}~flows} % x-y flows

\newcommand{\g}{\ensuremath{G\ }}
\newcommand{\G}{\ensuremath{G}}
\newcommand{\Gn}[1]{\ensuremath{G[#1]_n}}
\newcommand{\Gi}[1]{\ensuremath{G[#1]_i}}

\newcommand{\s}{s}

\newcommand{\half}{\frac{1}{2}}

\newcommand{\ceil}[1]{\ensuremath{\left\lceil #1 \right\rceil}}
\newcommand{\floor}[1]{\ensuremath{\left\lfloor #1 \right\rfloor}}

%----------------------
% lTOP
%----------------------

\newcommand{\ltp}{\ensuremath{|G|_\ell}}
\newcommand{\ltop}{\ensuremath{\ell-TOP}}
\newcommand{\blg}{\ensuremath{\partial^\ell G}}
\newcommand{\lER}{\ensuremath{\ell: E(G) \to \R_{>0}}}
\newcommand{\finl}{\ensuremath{\sum_{e \in E(G)} \ell(e) < \infty}}
\newcommand{\infl}{\ensuremath{\sum_{e \in E(G)} \ell(e) = \infty}}


%----------------------
% NETWORKS -- harmonic functions
%----------------------

\newcommand{\knl}{Kirchhoff's node law}
\newcommand{\kcl}{Kirchhoff's cycle law}
\newcommand{\cutr}{non-elusive} %cut respecting}
\newcommand{\Cutr}{Non-elusive} %cut respecting}

\newcommand{\are}{\vec{e}}
\newcommand{\arE}{\vec{E}}

\newcommand{\DP}{Dirichlet problem}
\newcommand{\BM}{Brownian Motion}
\newcommand{\hf}{harmonic function}

%----------------------
% RW --- PROBABILITY
%----------------------

\newcommand{\Ex}{\mathbb E}
\renewcommand{\Pr}{\mathbb{P}}

%----------------------
% REFERENCE
%----------------------

\newcommand{\Lr}[1]{Lemma~\ref{#1}}
\newcommand{\Lrs}[1]{Lemmas~\ref{#1}}
\newcommand{\Tr}[1]{Theorem~\ref{#1}}
\newcommand{\Trs}[1]{Theorems~\ref{#1}}
\newcommand{\Sr}[1]{Section~\ref{#1}}
\newcommand{\Srs}[1]{Sections~\ref{#1}}
\newcommand{\Prr}[1]{Pro\-position~\ref{#1}}
\newcommand{\Prb}[1]{Problem~\ref{#1}}
\newcommand{\Cr}[1]{Corollary~\ref{#1}}
\newcommand{\Cnr}[1]{Con\-jecture~\ref{#1}}
\newcommand{\Or}[1]{Observation~\ref{#1}}
\newcommand{\Er}[1]{Example~\ref{#1}}
\newcommand{\Dr}[1]{De\-fi\-nition~\ref{#1}}

%----------------------
% VERBAL SHORTCUTS
%----------------------

%graphs
\newcommand{\lf}{locally finite}
\newcommand{\lfg}{locally finite graph}
\newcommand{\fg}{finite graph}
\newcommand{\ig}{infinite graph}
\newcommand{\bos}{basic open set}
\newcommand{\Cg}{Cayley graph}
\newcommand{\nlf}{non-locally-finite}
\newcommand{\nlfg}{non-locally-finite graph}
\newcommand{\fhcy}{Hamilton cycle}
\newcommand{\hcy}{Hamilton circle}
\newcommand{\et}{Euler tour}
\newcommand{\ets}{Euler tours}

%quantifying etc.
\renewcommand{\iff}{if and only if}
\newcommand{\fe}{for every}
\newcommand{\Fe}{For every}
\newcommand{\fea}{for each}
\newcommand{\Fea}{For each}
\newcommand{\st}{such that}
\newcommand{\sot}{so that}
\newcommand{\ti}{there is}
\newcommand{\ta}{there are}
\newcommand{\tho}{there holds}
\newcommand{\wed}{well-defined}

\newcommand{\obda}{without loss of generality}
\newcommand{\btdo}{by the definition of}
\newcommand{\Btdo}{By the definition of}
\newcommand{\btco}{by the construction of}
\newcommand{\Btco}{By the construction of}
\newcommand{\wrt}{with respect to}
\newcommand{\fim}{finitely many}
\newcommand{\inm}{infinitely many}
\newcommand{\istc}{is straightforward to check}
\newcommand{\ises}{is easy to see}
\newcommand{\leth}{large enough that}

%common terms
\newcommand{\locon}{locally connected}
\newcommand{\pacon}{path connected}
\newcommand{\inhy}{induction hypothesis}

\newcommand{\FC}{Freudenthal compactification}
\newcommand{\tcs}{topological cycle space}
\newcommand{\tocir}{topological circle}
\newcommand{\topa}{topological path}
\newcommand{\tet}{topological Euler tour}

%Probability
\newcommand{\Rw}{Random walk}
\newcommand{\rw}{random walk}
\newcommand{\RW}{Random Walk}


%----------------------
% ENVIRONMENTS
%----------------------

\newcommand{\labequ}[2]{ \begin{equation} \label{#1} #2 \end{equation} } 
\newcommand{\labtequ}[2]{%\labtequc{#1}{#2}}
 \begin{equation} \label{#1} 	\begin{minipage}[c]{0.9\textwidth}  #2 \end{minipage} \ignorespacesafterend \end{equation} } 

\newcommand{\labtequc}[2]{ \begin{equation} \label{#1} 	\text{   #2 } \end{equation} } 

\newcommand{\labtequstar}[1]{ \begin{equation*}  	\begin{minipage}[c]{0.9\textwidth}  #1 \end{minipage} \ignorespacesafterend \end{equation*} }



\newcommand{\mymargin}[1]{% <- dieses % verhindert ein ungewolltes Leerzeichen
  \marginpar{%
    \begin{minipage}{\marginparwidth}\small%
      \begin{flushleft}%
        {\color{blue}#1}%
      \end{flushleft}%
   \end{minipage}%
  }%
}%
%:=, das vernuenftig aussieht
\newcommand{\assign}{
  \mathrel{\mathop{:}}=
}

\newcommand{\mySection}[2]{}

%&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&


\begin{document}
\begin{thebibliography}{10}

\bibitem{AhBeRai}
R.~Aharoni and E.~Berger.
\newblock Rainbow matchings in $r$-partite $r$-graphs.
\newblock {\em Electronic J.\ Comb.}, 16:{R119}, 2009.

\bibitem{alon_edge_2007}
{Noga} {Alon} and {Vera} {Asodi}.
\newblock Edge colouring with delays.
\newblock {\em Combinatorics, Probability and Computing}, 16(02):173--191,
  2007.

\bibitem{BrualdiRyser}
Richard~A. Brualdi and Herbert~J. Ryser.
\newblock {\em Combinatorial Matrix Theory}.
\newblock Cambridge University Press, July 1991.

\bibitem{DiestelBook05}
R.~Diestel.
\newblock {\em Graph {T}heory \emph{(3rd edition)}}.
\newblock Springer-Verlag, 2005.
\newblock Electronic edition available at: \url{http://diestel-graph-theory.com/index.html}.

\bibitem{hallAbelian}
Marshall Hall~Jr.
\newblock A combinatorial problem on abelian groups.
\newblock {\em Proceedings of the American Mathematical Society},
  3(4):584--587, 1952.

\bibitem{HaWiWi}
G.T.~Wilfong P.E.~Haxell and P.~Winkler.
\newblock Delay coloring and optical networks.
\newblock Preprint (2001).

\bibitem{ryser}
H.J. Ryser.
\newblock Neuere {P}robleme in der {K}ombinatorik.
\newblock In {\em Vortr\"age \"uber {K}ombinatorik, {O}berwolfach}, pages
  69--91, 1967.

\bibitem{SteTra}
S.~K. Stein.
\newblock Transversals of Latin squares and their generalizations.
\newblock {\em Pacific Journal of Mathematics}, 59(2):567--575, 1975.

\end{thebibliography}

\maketitle


\begin{abstract}
In this note we prove the conjecture of  Wilfong, Haxell and Winkler (2001) that every bipartite multigraph with integer edge delays admits an edge colouring with $d+1$ colours in the special case where $d=3$. A connection to the Brualdi-Ryser-Stein conjecture is discussed.
\end{abstract}

\section{Introduction}

Motivated by scheduling issues in optical networks, Wilfong Haxell and Winkler \cite{HaWiWi} made the following elegant combinatorial conjecture:

\begin{conjecture} \label{conjdelay}
Let \g be a bipartite multigraph with partition classes $A,B$ and maximum degree $d$. Suppose that each edge $e$ is associated with an integer `delay' $r(e)$. Then $G$ admits an edge-colouring $f: E(G)\to \{0,\ldots,d\}$ \st\ $f$ is proper on $A$ and $f+r(mod\ d+1)$ is proper on $B$.
\end{conjecture}

See also \cite{alon_edge_2007}.
When the graph consists of just two vertices joined by $d$ parallel edges, this is implied by a theorem of Hall~\cite{hallAbelian} as noted in \cite{HaWiWi}. This fact has been dubbed `the fundamental theorem of juggling' (think of the two vertices as the two hands of a juggler juggling $d$ balls).

More generally, one can consider the case where the edges impose an arbitrary distortion of the colours, given by a permutation $r$ of the set of colours, rather than `delaying' the colour by a constant:

\begin{problem} \label{p}
Let \g be a bipartite multigraph with partition classes $A,B$ and maximum degree $d$. Suppose that each edge $e$ is associated with a permutation $r_e$ of the set $\{0,1,\ldots, d\}$. Then $G$ admits a proper distortion-colouring with colours $\{0,1,\ldots, d\}$ (definitions in the next section). 
%n edge-colouring $f: E(G)\to \{0,\ldots,d\}$ \st\ $f$ is proper on $A$ and the colouring $e\mapsto p_e(f(e))$ is proper on $B$.
\end{problem}

It turns out that in this generality the problem becomes much harder than its special case in \Cnr{conjdelay}: as noticed by N.~Alon (private communication), the special case of \Prb{p} where each class $A,B$ consists of just one vertex is equivalent to the following conjecture of \cite{AhBeRai}, which is a strengthening of a well-known conjecture of Brualdi and Stein on transversals in Latin squares.

\begin{conjecture}[{\cite[Conjecture 2.4\footnote{The conjecture of \cite{AhBeRai} is more general than \Cnr{a}: it allows for larger sets $V_2,V_3$ with an appropriate modification of the perfect matching condition.}]{AhBeRai}}] \label{a}
Let $H$ be a 3-partite 3-uniform hypergraph, with partition classes $V_1,V_2,V_3$ \st\ $|V_2| =|V_3|=|V_1| + 1$. Suppose that for every $x \in V_1$, the set of hyperedges containing $x$ induces a perfect matching of $V_2 \cup V_3$. Then $V_1$ is matchable.
\end{conjecture}
Here, $V_1$ being matchable means that there is a matching in $H$ containing all vertices in $V_1$.

Indeed, to see the equivalence, represent each edge in \Prb{p} by a vertex in $V_1$, and let $V_2, V_3$ be sets of size $d+1=|V_1|+1$, to be thought of as the colour on the left endvertex and the colour on the right endvertex respectively.

\Cnr{a} strengthens the following well-known conjecture, made independently by  Brualdi \cite{BrualdiRyser} and Stein \cite{SteTra}

%of Brualdi that every latin square of odd size $n$ admits a  transversal of $n-1$ elements.

\begin{conjecture}[Brualdi-Stein] \label{latin}
In every $n \times n$ Latin square there exists a transversal of size $n-1$.
\end{conjecture}
An $n \times n$ matrix with entries in $\{1,\ldots, n\}$ is a \defi{Latin square}, if no two entries in the same row or in the same column are equal. A \defi{transversal} is a set of entries, each in a different row and in a different column, and each containing a different symbol. (For odd $n$, Ryser \cite{ryser} conjectured that there is even a transversal of size $n$.)

To see that \Cnr{a} implies \Cnr{latin}, construct a 3-partite hypergraph $H$ with $V_1$ being the set of rows of that Latin square $L$, $V_2$ the set of columns, and $V_3= \{1,\ldots, n\}$, and for each entry of $L$ introduce an edge containing the three corresponding vertices of $H$. Then delete an arbitrary vertex in $V_1$ with all edges containing it to obtain the setup of \Cnr{a}.

\medskip

The aim of this note is to prove that \Prb{p} has a positive answer for $d=3$.


%We prove  \cite[Conjecture 1.1]{HaWiWi} ---see also \cite{alon_edge_2007}--- for $d=3$ in a more general form, allowing the edges to `distort' the colours by an arbitrary bijection rather than by addition of a constant delay.

\section{Definitions}


Let $G=(V,E)$ be a bipartite multigraph of maximum degree $d$ with bipartition $\{A,B\}$. Let  $Col:= \{0,1,\ldots, d\}$ be the set of \defi{colours}, and suppose that every edge $e\in E$ is associated with  a bijection $r_e$ on $Col$, called the \defi{distortion} of $e$; intuitively, we are going to colour $e$ at its $A$ end and the colour will be distorted by $r_e$ when seen from $B$ (in \cite{HaWiWi} $r_e$ was addition,  mod $d+1$,  with the `delay' of $e$). If $a,b$ is the endvertex of $e$ in $A,B$ respectively, then we use the notation $ba(\cdot)$ to denote $r_e(\cdot)$ and $ab(\cdot)$ to denote $r^{-1}_e(\cdot)$.

A $k$-colouring of a set $E$ (of edges) is a function $f : E \to \{0,1,\ldots, k-1\}$. 
A $k$-colouring $f$ of the edges of a multigraph $G$ as above is called a \defi{proper distortion-colouring} (\wrt\ the permutations $r_e$), if \fe\ $a\in A$ we have $f(e)\neq f(g)$ \fe\ two edges $e,g$ incident with $a$ and \fe\ $b\in B$ we have $r_e(f(e))\neq r_g(f(g))$ \fe\ two edges $e,g$ incident with $b$.

\section{Main}

\begin{theorem}
\Fe\ bipartite multigraph \g of maximum degree 3, and any edge distortions, 
\ti\ a proper distortion-colouring of $E(G)$ with 4 colours 0,1,2,3.
\end{theorem}
\begin{proof}
We may assume without loss of generality that every vertex of \g has degree precisely 3, for otherwise we can add dummy vertices and edges to make \g cubic. It is well known that the edges of a regular bipartite multigraph can be decomposed into disjoint perfect matchings \cite[Corollary 2.1.3]{DiestelBook05}. So let $M,M',M''$ be perfect matchings of \g with $M \cup M' \cup M''=E(G)$. Then $M' \cup M''$ is a 2-factor, and it can be decomposed into a collection \cc\ of edge disjoint cycles.

Let $\{A,B\}$ be the bipartition of $V(G)$. We are going to let each element of \cc\ choose the colours of the edges of $M$ incident with its $A$ side. More precisely, given a $C \in \cc$, let $M_{C\cap A}$ denote the set of edges in $M$ incident with $C\cap A$, and let $M_{C\cap B}$  denote the set of edges in $M \sm M_{C\cap A}$ incident with $B$. We are going to prove that 
%
\labtequ{MA}{\Fe\ $C \in \cc$, \ti\ a 4-colouring $f_A$ of $M_{C\cap A}$ \st\ \fe\ 4-colouring $f_B$ of $M_{C\cap B}$, there is a 4-colouring $f_C$ of $E(C)$ \st\ $f_A \cup f_B \cup f_C$  is a proper distortion-4-colouring.}
%
Note that \eqref{MA} easily implies a proper distortion-colouring of $E(G)$ with 4 colours: the sets $M_{C\cap A}\mid C\in\cc$ are pairwise edge disjoint, and their union is $M$. Thus, we can begin by colouring each of them by a colouring $f_A$ as in \eqref{MA}, and then we can extend the colouring to each $C\in\cc$ keeping it proper.

So let us prove \eqref{MA}. Given a $C \in \cc$, pick a 2-edge subarc $uvy$ of $C$ with $u,y\in A$. Distinguish two cases:

If the distortions of the edges $uv,vy$ are identical, then give the edges $m_u,m_y$ of $M$ incident with $u,y$ colours that are different (when seen from $A$). If $C$ happens to be a 2-cycle, in which case $u=y$, give $m_u=m_y$ any colour.

If those distortions are not identical, then colour (the $A$ side of) both $m_u,m_y$  with a colour $\alpha$  \st\ $vu(\alpha)\neq vy(\alpha)$. 

In both cases, colour the rest of $M_{C\cap A}$ arbitrarily; those colours will not matter.

We claim that this colouring $f_A$ has the desired property. To prove this, let $f_B$ be any colouring of $M_{C\cap B}$, and note that \fe\ edge $e\in E(C)$ the set $L_e$ of still available colours for $e$, that is, the colours that would not conflict with $f_A \cup f_B$ if given to $e$ on its $B$ side, say, has at least 2 elements; indeed, only 2 edges adjacent with $e$ have been coloured so far and we had 4 colours to begin with. 

Let us first deal with the case where $C$ is not a 2-cycle, and  consider again the two edges $vu,vy$ as above.
We claim that $L_{vu}\neq L_{vy}$; indeed, the colours we gave to $m_u,m_y$ were chosen is such a way so as to forbid a different candidate colour at the $v$ side of each of $vu,vy$, and so $L_{vu}\neq L_{vy}$ holds. 
On the other hand, the colour of the  edge in $M$ incident with $v$ forbids the same colour for each of $vu,vy$, which implies that $L_{vu}\cap L_{vy}\neq \emptyset$. 

Thus, since $L_{vu}, L_{vy}$ are neither equal nor disjoint, and each contains at least two colours, we can find a common colour $\beta \in L_{vu}\cap L_{vy}$ and another 2 colours $\gamma\in L_{vu}, \delta\in L_{vy}$ so that $\beta, \gamma,\delta$ are all distinct. Now colour $vu$ with $uv(\gamma)$ (so that its colour seems to be $\gamma$ on its $B$ side),
% and $vy$ with $\delta$. 
and note that our colouring is still proper, since this colour came from the allowed list. Consider the next edge $ux$ of $C$ incident with $u$. This edge still has at least 1 available colour after we coloured $uv$ (recall that $|L_e|\geq 2)$, so give it that colour. Continue like this along $C$, to properly colour all its edges except  the last edge $vy$. Now note that when we coloured $vu$ we still left 2 colours available for the $B$ side of $vy$, namely $\beta,\delta\neq \gamma$. At least one of them is still available now, and we assign it to the $B$ side of $vy$ completing the proper distortion-colouring of $C$.

If $C$ is a 2-cycle then the situation is much simpler, and it is straightforward to check that \eqref{MA} holds by distinguishing two cases according to whether its 2 edges bear the same distortions.

This completes the proof. Note that we proved something stronger than \eqref{MA}: for each $C \in \cc$, all but one of the edges in $M_{C\cap A}$ can be precoloured arbitrarily. 
\end{proof}

\comment{
\begin{problem}
Prove that \cite[Conjecture 1.1]{HaWiWi} is true \iff\ it is true for arbitrary distortions instead of delays.
\end{problem}
}



\end{document}